\begin{table}%ta2
 %%\centering

\par
\caption{\label{tbl:sed}Literature photometry of HD~46149.}
\begin{tabular}{rllr}
\hline\hline\\[-3mm]
Wavelength (\AA) & System & Band & Reference\\\hline
  900--1150 &   FUSE &     Far-UV &  \citet{fuse}\\
 1150--3300 &    IUE &         UV &  \citet{iue}\\
 2740 &        TD1 &         UV &  \citet{TD1}\\
 3641 &    JOHNSON &          $U$ &  \citet{johnson}\\
 4448 &    JOHNSON &          $B$ &  \citet{johnson}\\
 5504 &    JOHNSON &          $V$ &  \citet{johnson}\\
12~487 &    JOHNSON &          $J$ &  \citet{johnson}\\
16~464 &    JOHNSON &          $H$ &  \citet{johnson}\\
21~951 &    JOHNSON &          $K$ &  \citet{johnson}\\
 4204 &      TYCHO &         BT &  \citet{tycho}\\
 5321 &      TYCHO &         VT &  \citet{tycho}\\
 4718 &       SDSS &          $g$ &  \citet{sdss}\\
 6185 &       SDSS &          $r$ &  \citet{sdss}\\
 7499 &       SDSS &          $i$ &  \citet{sdss}\\
 8961 &       SDSS &          $z$ &  \citet{sdss}\\
 7885 &    COUSINS &          $I$ &  \citet{cousins}\\
12~412 &      2MASS &          $J$ &  \citet{2mass}\\
16~497 &      2MASS &          $H$ &  \citet{2mass}\\
21~909 &      2MASS &         $K_{\rm S}$ &  \citet{2mass}\\
35~375 &       IRAC &         36 $\mu$m &  \citet{irac}\\
44~750 &       IRAC &         45 $\mu$m &  \citet{irac}\\
57~019 &       IRAC &         58 $\mu$m &  \citet{irac}\\
77~843 &       IRAC &         80 $\mu$m &  \citet{irac}\\
87~973 &        MSX &          $A$ &  \citet{msx}\\\hline
\end{tabular}
\end{table}